

<sup>a</sup>Department of Orthopedic Surgery, <sup>d</sup>Department of Laboratory Medicine and Pathology, <sup>e</sup>Department of Health Sciences Research, <sup>†</sup>Department of Physical Medicine and Rehabilitation, <sup>g</sup>Division of Pain Medicine, Department of Anesthesiology, <sup>h</sup>Department of Radiology, <sup>m</sup>Department of Anatomy, and <sup>n</sup>Department of **Biochemistry and Molecular** Biology, Mayo Clinic College of Medicine, Mayo Clinic, Rochester, Minnesota, USA; <sup>c</sup>Department of Biomedical Engineering and Physiology, Mayo Graduate School, Rochester, Minnesota, USA; <sup>i</sup>Department of Biochemistry, <sup>j</sup>Department of Orthopedic Surgery, and <sup>k</sup>Section of Rheumatology, Department of Internal Medicine, Rush University Medical Center, Chicago, Illinois, USA; <sup>I</sup>Jesse **Brown Veterans Affairs** Medical Center, Chicago, Illinois, USA; <sup>b</sup>Department of Orthopedic Surgery, Tongji Hospital, Tongji Medical College, Huazhong University of Science and Technology, Wuhan, People's Republic of China

#### \*Contributed equally.

Correspondence: Andre van Wijnen, Ph.D., Department of Orthopedic Surgery, Medical Sciences Building, Room S3-69, Mayo Clinic, 200 1st Street SW, Rochester, Minnesota, 55905, USA. Telephone: 507-773-2105; e-mail: vanwijnen.andre@mayo. edu

Received February 22, 2016; accepted for publication September 1, 2016; published Online First on October 26, 2016. ©AlphaMed Press 1066-5099/2016/\$20.00/0

#### http://dx.doi.org/ 10.5966/sctm.2016-0097

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

## Safety Studies for Use of Adipose Tissue-Derived Mesenchymal Stromal/Stem Cells in a Rabbit Model for Osteoarthritis to Support a Phase I Clinical Trial

Scott M. Riester,<sup>a,\*</sup> Janet M. Denbeigh,<sup>a,\*</sup> Yang Lin,<sup>a,b</sup> Dakota L. Jones,<sup>a,c</sup> Tristan de Mooij,<sup>a</sup> Eric A. Lewallen,<sup>a</sup> Hai Nie,<sup>a</sup> Christopher R. Paradise,<sup>a</sup> Darcie J. Radel,<sup>d</sup> Amel Dudakovic,<sup>a</sup> Emily T. Camilleri,<sup>a</sup> Dirk R. Larson,<sup>e</sup> Wenchun Qu,<sup>fg</sup> Aaron J. Krych,<sup>a</sup> Matthew A. Frick,<sup>h</sup> Hee-Jeong Im,<sup>i,j,k,I</sup> Allan B. Dietz,<sup>d</sup> Jay Smith,<sup>f,h,m</sup> Andre J. van Wijnen<sup>a,c,n</sup>

Key Words. Cell therapy • Stem cell • Transplant • Regenerative medicine • Osteoarthritis • Rabbit

## ABSTRACT

Adipose-derived mesenchymal stem cells (AMSCs) offer potential as a therapeutic option for clinical applications in musculoskeletal regenerative medicine because of their immunomodulatory functions and capacity for trilineage differentiation. In preparation for a phase I clinical trial using AMSCs to treat patients with osteoarthritis, we carried out preclinical studies to assess the safety of human AMSCs within the intra-articular joint space. Culture-expanded human AMSCs grown in human platelet-lysate were delivered via intra-articular injections into normal healthy rabbit knees and knees at risk for the development of osteoarthritis after bilateral medial anterior hemimeniscectomy. Treatment outcomes and safety were evaluated by assessing the general health, function, and behavior of the animals. Joint tissues were analyzed by x-ray, magnetic resonance imaging, and histopathology. Intra-articular AMSC therapy was well tolerated in this study. We did not observe adverse systemic reactions, nor did we find evidence of damage to intra-articular joint tissues. Thus, the data generated in this study show a favorable safety profile for AMSCs to treat osteoarthritis. Stem Cells Translational *MEDICINE 2017;6:910–922* 

## SIGNIFICANCE STATEMENT

This safety study indicates that adipose-derived mesenchymal stem cells (AMSCs) can be safely used in cell therapies for joint degeneration. This finding supports a phase I clinical trial to evaluate the clinical utility of AMSCs in treating osteoarthritis.

## INTRODUCTION

Osteoarthritis (OA) is a major musculoskeletal degenerative disease that causes significant physical disability and pain. It is estimated that approximately 7 million Americans are currently living with a hip or knee replacement secondary to osteoarthritis-related joint degeneration [1]. By 2030, in the U.S. alone, there are projected to be almost 4 million primary total knee or hip arthroplasties performed annually for osteoarthritisrelated joint degeneration [2]. Articular cartilage has a reduced innate regenerative capacity [3]; however, a number of clinical therapies are commonly used for OA treatment. These include oral nonsteroidal anti-inflammatory medications, viscosupplementation (i.e., SYNVISC and Supartz), platelet-rich plasma, and intra-articular corticosteroid therapies [4-7]. Intra-articular corticosteroid injections have been used for decades to treat osteoarthritis pain symptoms, and they currently serve as the primary treatment alternative to joint replacement for end-stage OA. Although effective at relieving acute pain symptoms, their use over time may contribute to cartilage breakdown and accelerated progression of osteoarthritis [8]. These current treatments are primarily aimed at symptomatic relief of pain symptoms and are associated with significant potential for adverse side effects. To date, there are no therapies in clinical use that have been shown to clearly slow or reverse osteoarthritis progression.

Adipose-tissue derived mesenchymal stromal/ stem cells (AMSCs) are immature mesenchymal cells with stem cell-like properties that are derived from the stromal vascular fraction of adipose tissue. AMSCs are the focus of our studies because they can be more easily obtained—and in vastly greater quantities—than mesenchymal stromal cells (MSCs) from other sources. As such, AMSCs are amenable to therapeutic applications that require or would benefit from autologous cell transplants. Furthermore, AMSCs expanded in human platelet lysate are zoonotic-free cells that have promising potential for clinical orthopedic applications, based on their ability to differentiate into different types of mesenchymal cells, including those of the osteogenic, chondrogenic, and adipogenic lineages [9-12]. In addition to their capacity for multilineage differentiation, these cells may support tissue regeneration via their trophic and secretory functions, including anti-inflammatory activity [13, 14]. During OA pathogenesis, joint inflammation promotes a catabolic state characterized by the loss of extracellular matrix, in which chondrocytes are susceptible to cell death and apoptosis [15]. The antiinflammatory activity of AMSCs may relieve pain symptoms of OA without the risk of cartilage degeneration that can occur with corticosteroid use.

In this study, we address the important question of whether platelet-lysate expanded AMSCs can be safely used within the intra-articular joint space. We examine the biological effects of injecting human AMSCs into the intra-articular knee space of normal healthy knees (cohort A) and of knees at risk for the development of osteoarthritis secondary to medial anterior hemimeniscectomy (cohort B) using a translational rabbit model for OA. The main finding of our study is that AMSCs are well-tolerated in both models, even though we observed minor inflammatory responses that are attributable to cell xenografts in an immunocompetent animal. This study provides preliminary safety data in support of an approved investigational new drug application to the Food and Drug Administration for initiation of a phase I clinical trial for in human use of autologous AMSCs to treat osteoarthritis.

#### MATERIALS AND METHODS

#### **Overview of the Study Design**

The safety profile of human AMSCs was evaluated in two cohorts of mature female New Zealand white rabbits (12 months old, body weight 2.5-3.5 kg) (Harlan Laboratories/Envigo, Oxford, MI, http://www.envigo.com) In the first group (cohort A, n = 13), healthy rabbits received intra-articular knee injections of AMSCs at a dose of either  $2 \times 10^6$  or  $12 \times 10^6$  cells or lactated Ringer's control into their right knee. Investigators performing the injections and evaluating the rabbit vitals and tissue histology were blinded to the type of experimental treatment being administered. Additionally, 1 ml of lactated Ringer's control solution was also injected into the left knee of each rabbit as an internal control. Rabbits were sacrificed at either 7 days  $(2 \times 10^6 \text{ cells}; n = 2; 12 \times 10^6 \text{ cells}; n = 2; \text{ lactated Ringer's con-}$ trol: n = 3) or 14 days (2 × 10<sup>6</sup> cells: n = 2; 12 × 10<sup>6</sup> cells: n = 2; lactated Ringer's control: n = 2) after experimental injection (Fig. 1). Safety and treatment outcomes were evaluated by daily measurement of rectal temperature and body weight, as well as visual inspection of the injection site and knee joint. Joint tissues were fixed and histologically evaluated at the completion of the study.

Because our ultimate goal is to use AMSCs to treat OA in patients, we also examined the safety of AMSCs in a second group of rabbits at risk for osteoarthritis development secondary to partial medial anterior hemimeniscectomy. These rabbits (cohort B; n =18) underwent bilateral medial anterior hemimeniscectomies, followed by experimental injections of either AMSCs at a dose of  $2 \times 10^6$  cells (n = 6) or  $6 \times 10^6$  cells (n = 6) or lactated Ringer's control (n = 6) into their right knee 4 weeks later (Fig. 1). Treatment outcomes and safety were evaluated by assessing weekly rectal temperature and body weight. The rabbit's surgical incision and injection sites were also visually inspected daily. Additionally, the effect of AMSC therapy on intra-articular joint structures was assessed by using x-ray, magnetic resonance (MR) imaging, and histologic evaluation of joint tissues. Before animal sacrifice, a pain evaluation was performed by using weight-bearing measures (incapacitance testing). All animal studies were carried out with approval from the Mayo Clinic's Institutional Animal Care and Use Committee.

#### Primary Culture of Human AMSCs

Human AMSCs were cultured from adipose tissue collected from patients undergoing elective liposuction surgery in compliance with protocols approved by the Mayo Clinic's institutional review board. For specific details about human AMSC harvesting, and in vitro culturing, refer to Crespo-Diaz et al. [11]. In brief, AMSCs were cultured in advanced minimum essential medium (Thermo Fisher Scientific Life Sciences, Oakwood Village, OH, https://www.thermofisher.com) supplemented with 5% (vol/ vol) human platelet lysate (PLTMax, Mill Creek Life Sciences, Rochester, MN, http://www.millcreekls.com), 1% (vol/vol) Penn-Strep (100 U/ml penicillin and 100  $\mu$ g/ml streptomycin; Corning, Corning, NY, http://www.corning.com), 2 mM Glutamax (Thermo Fisher), and 2 U/ml heparin (Vizient, Irving, TX, https://www.vizientinc.com). Standard operating procedures were followed according to good manufacturing practice guidelines. Passage 5 cells were used for the injection procedures to ensure that a sufficient number of cells would be available for both molecular testing and animal injection studies. Additionally, AMSCs have been shown to maintain their proliferative potential at later passages when grown in human platelet lysate, compared with adult-derived stem cells grown in fetal bovine serum [11]. Stem cell populations were confirmed by using established criteria from the International Society for Cellular Therapy and the International Federation for Adipose Therapeutics and Science [16]. All cells were characterized according to their cell surface profile and were positive for CD44, CD73, CD90, CD105, and HLA-ABC and negative for CD14, HLA-DR, and CD45. These cell surface markers have previously been validated as markers that define an adipose-derived stem cell population with trilineage potential [9-11]. At the time of cell preparation, AMSCs were detached from flasks by using TrypLE Express (Thermo Fisher) and counted by using a hemocytometer (Thermo Fisher). Cells were suspended in 1 ml of lactated Ringer's solution at concentrations of (a) 2 imes 10<sup>6</sup> or (b) 12 imes10<sup>6</sup> cells per milliliter for cohort A and at concentrations of (c)  $2 \times 10^6$  or (d)  $6 \times 10^6$  cells per milliliter for cohort B. Cell suspensions were stored on wet ice until ready for administration. The higher cell dosing regimen was used in cohort A to evaluate for potential acute adverse reactions, including systemic and localized immune reactions. For cohort B, we used lesser doses to simulate clinically relevant scenarios. We used doses that were similar to those used in previous studies when accounting for scaling differences between human and rabbit knees [17], because some studies have suggested that highdose therapy (doses greater than 150 million cells injected into a human knee) may be less efficacious [18].

© 2016 The Authors



## COHORT A



**Figure 1.** Study design. An experimental timeline showing the procedures for cohorts A and B. Top: Cohort A: Injection of right and left rabbit knees at time zero (left) followed by sacrifice and tissue harvest at either 7 days after injection (center) or 14 days after injection (right). The left knee of each rabbit was injected with lactated Ringer's while the right knees were treated with either lactated Ringer's,  $2 \times 10^6$  cells, or  $12 \times 10^6$  cells. Bottom: Cohort B: Injection of right and left rabbit knees at 4 weeks after bilateral anterior hemimeniscectomies. The left knee of each rabbit was injected with lactated Ringer's, while the right knees were treated with lactated Ringer's,  $2 \times 10^6$  cells, or  $6 \times 10^6$  cells. Animals in this cohort were sacrificed 4 weeks after injection (8 weeks after partial meniscectomy).  $t_i$  One rabbit in cohort A receiving lactated Ringer's injection into its right knee was sacrificed before receiving injection therapy because of post-operative complications related to the surgical meniscectomy procedure. Abbreviation: R, right.

## **Rabbit Meniscectomy**

To increase the susceptibility of the rabbit knees to osteoarthritis development, we performed bilateral medial anterior hemimeniscectomies on all rabbits in cohort B, before AMSC injection (Fig. 2). The development of OA is generally characterized by macroscopically visible surface fibrillation and focal erosions/lesions that increase in number and severity to 12 weeks and beyond [19–23], as well as osteophyte development [24].

Anesthesia was carried out via administration of an intramuscular injection of ketamine (35–75 mg/kg) and 5 mg of xylazine, with inhalation of isoflurane (1%–4%). Rabbits were intubated with a 2.5–3.5 endotracheal tube and maintained on 100% oxygen and isoflurane for the procedure. Buprenorphine SR 0.18 mg/kg (subcutaneous; s.c.) was given for pre-emptive analgesia, as well as cefazolin (22 mg/kg) i.v. to prevent infections. Body temperature was monitored by using a rectal thermometer, with a recirculating warm water pad used as necessary. Once anesthetized, electric clippers were used to remove the hair overlying each knee, and the skin was prepared by using Hibiclens (Mölnlycke Health Care U.S., Norcross, GA, http://www.molnlycke.us). Sterile techniques were used during the operation. A paramedian vertical skin incision (1–2 cm) was made parallel to the patellar tendon, and an arthrotomy was performed. The anterior insertional ligament of the medial meniscus was released, and the medial aspect of the meniscus was dissected free from its attachment at the medial collateral ligament. The joint was irrigated with normal saline, and the wound was closed in layers, using 3-0 Vicryl interrupted simple sutures for the fascia and the ethylon 5-0 interrupted horizontal mattress suture for the skin (Fig. 2). Carprofen 4 mg/kg s.c. was given at the time of surgery and once every 24 hours for 3 days postoperatively for pain control.

After surgery, rabbits were placed back into a holding pen, where they were allowed to recover; they were not transported until they had regained mobility. Every animal was physically examined twice within the first 24 hours. One rabbit in cohort B was euthanized because of an infection of the left knee after meniscectomy, before the injection of AMSCs. All other animals in this cohort recovered normally and underwent a single bilateral intra-articular knee injection of human AMSCs 4 weeks after the meniscectomy.

### © 2016 The Authors

STEM CELLS TRANSLATIONAL MEDICINE



Figure 2. Rabbit meniscectomy. Surgical procedure for medial anterior hemimeniscectomy. (A): Rabbit knee after shaving and anatomical marking. (B): A vertical incision made along the medial aspect of the knee joint reveals the patellar tendon and medial collateral ligament. (C): An incision is made along the joint line to expose the meniscus. (D): The infrapatella fat pad is retracted, exposing the lateral insertion of the medial meniscus. (E): The lateral insertion site is released. (F): The meniscus is reflected medially out of the joint. (G): Its attachment at the medial collateral ligament is released. (H): The resected medial meniscus. (I): The fascia and skin are then closed with sutures. (J): The rabbit knee after surgery.

## **Intra-Articular Rabbit Knee Injections**

Before intra-articular injection of either AMSCs or lactated Ringer's treatment, each rabbit was anesthetized with an intramuscular injection of ketamine (35 mg/kg) and xylazine (5 mg/kg), intubated, and prepared as described above for the meniscectomy procedure. A 3-ml plastic syringe (BD Biosciences, San Jose, CA, http://www.bdbiosciences.com) was loaded with 1 ml of lactated Ringer's solution either with or without a cell suspension, to

which a 19-gauge stainless steel needle (Medtronic, Minneapolis, MN, http://www.medtronic.com) was subsequently attached. The needle was inserted into the knee joint, and the experimental therapy was delivered under ultrasound guidance (CX50 ultrasound machine, Linear 12-3 MHz probe, Philips Healthcare, Fitchburg, WI, http://www.usa.philips.com) with the use of sterile ultrasound gel to ensure appropriate intra-articular delivery. In both study cohorts, ultrasound imaging was used to identify

© 2016 The Authors



**Figure 3.** Injection of AMSCs into anesthetized rabbits. Intra-articular knee injection procedure after shaving, with the injection of stem cells into the rabbit knee joint under ultrasound image guidance, is shown. **(A)**: The rabbit is anesthetized and positioned in the supine position on the operating table. **(B)**: A 19-gauge needle is inserted into the intra-articular space along the joint line under ultrasound guidance. **(C)**: Ultrasound imaging shows the needle (arrowhead) passing between the patellar tendon and the femur positioned within the trochlear groove. Abbreviations: F, femur; PT, patellar tendon.

anatomic landmarks within the knee joint (Fig. 3). The needle was inserted into the knee joint between the patellar tendon and trochlea by using a single pass, in plane approach to visualize the shaft and tip of the needle. Special care was taken to avoid iatrogenic cartilage injury during the injection. The injectate was observed to distend the suprapatellar recess, indicating successful delivery of the experimental therapy into the knee. After injection, the knee was placed through several ranges of motion and re-examined with ultrasound (US) to confirm joint distention. At the completion of the procedure, each animal received 120 ml of PlasmaLyte (Baxter, Utrecht, The Netherlands, http://www. baxter.nl) s.c. for recovery; no antibiotics or analgesics were administered.

## Pain Evaluation (Incapacitance Testing)

To determine whether rabbits experienced measurable differences in discomfort or pain sensation between knees treated with AMSCs and lactated Ringer's control, the weight distribution on the two hind limbs of rabbits in cohort B (at 8 weeks after meniscectomy) were measured by using an incapacitance meter (Incap, Columbus Instruments, Columbus, OH, http://www.colinst.com). Alert rabbits were measured in a neutral (semiupright; using the manufacturer-provided animal restrainer) or elevated (forelimbs lifted while eyes covered) position with hind paws placed on separate sensor pads, and the force exerted by each limb (measured in grams) averaged over 5 seconds per reading. To minimize distress during testing, the eyes of each rabbit were covered. Incapacitance measurements for each rabbit were taken three times in both the neutral and semiupright positions.

## **Animal Sacrifice**

At the completion of the study, rabbits were euthanized either 1 or 2 weeks after injection (cohort A) to evaluate for acute adverse reactions and at 4 weeks after injection (cohort B) to evaluate for chronic adverse effects on cartilage tissues. Euthanization was carried out via intracardiac administration of 3 ml of Fatal Plus

STEM CELLS TRANSLATIONAL MEDICINE published by Wiley Periodicals, Inc. on behalf of AlphaMed Press

(392 mg/ml pentobarbital sodium) (Vortech Pharmaceuticals, Ltd., Dearborn, MI) after a ketamine (35 mg/kg) and xylazine (5 mg/kg) injection. Immediately after sacrifice, the animals in cohort B underwent x-ray and MR imaging before tissue harvest.

## X-Ray Imaging

The hind limbs of animals in cohort B were imaged immediately after surgery, at 4 weeks (after anesthetization), and at 8 weeks (immediately after sacrifice) postoperation by using anteriorposterior (AP) and lateral x-rays to look for signs of OA or other joint damage (e.g., osteophytes). X-rays of both knees were taken by using the HF 100\30+ Ultra light (MinXray Inc., Northbrook, IL, http://www.minxray.com). X-ray radiographs were then digitized (ScanX, ALLPRO Imaging, Melville, NY, http://allpro-imaging. com). Radiographs were evaluated by a trained orthopedic surgeon, who was blinded to the treatment type administered in the right knee. X-rays were evaluated for signs of OA and joint destruction through assessment of osteophyte formation, jointspace narrowing, sclerosis, and joint deformity.

## **Magnetic Resonance Imaging**

Magnetic resonance imaging with T2-weighting was performed on rabbit hind limbs in cohort B to assess potential pathological changes in the articular cartilage of the knees [25, 26]. After sacrifice and x-ray imaging, both rabbit legs were disarticulated and prepared for MR scanning. The legs were bound in the fully extended position and placed between custom manufactured paired 2-channel receive only phased array coils (Mayo Clinic MR Laboratory, Rochester MN, http://www.mayoclinic.org). Imaging was performed on a 3.0-T MRI (Signa, GE Healthcare Life Sciences, Piscataway, NJ, http://www.gelifesciences.com). Fast-spin echo axial fat-saturated T2-weighted images and matched sagittal proton density and fat-saturated T2-weighted images were obtained through both knees for each rabbit specimen using the following parameters: field of view =  $8 \times 8$  cm; repetition time = 4,000 milliseconds; echo time = 40–60 milliseconds; matrix =  $512 \times 320$  with

© 2016 The Authors

Stem Cells Translational Medicine

2-mm slice thickness, no interslice gap, and number of excitations = 2. T2-weighted maps were generated for each rabbit specimen by using an Advantage Windows workstation, platform 4.3 (GE Health-care). The average T2 relaxation times for the articular cartilage corresponding to the medial and lateral femoral condyles and medial and lateral tibial plateau of both the left and right knees of each animal were measured within a region of interest (4 mm<sup>2</sup>).

#### **Tissue Histology**

Joint tissues were harvested immediately after sacrifice in cohort A and after x-ray and MR imaging in cohort B. The distal femur (cohort A only), proximal tibia, lateral meniscus, synovium, and fat pad were fixed in 10% neutral buffered formalin for 72 hours in preparation for paraffin embedding. The articular surface and associated subchondral bone from the proximal tibia and distal femur were sectioned into four quadrants and decalcified by using the RDO Rapid Decalcifying Solution (Apex Engineering Products Corp., Aurora, IL, http://www.apexengineeringproducts.com) for several hours until the bone tissue was completely decalcified. Tissue specimens were embedded in paraffin and cut into 5- $\mu$ m sections for staining. All specimens were mounted onto slides and stained using protocols for hematoxylin and eosin (H&E) staining. In addition, all cohort B tissues from rabbits receiving injections of either lactated Ringer's or 6 imes 10<sup>6</sup> AMSCs into their right (experimental) knees were also stained for safranin O and counterstained with fast green so that Osteoarthritis Research Society International (OARSI) scores could be evaluated [26, 27]. Histological sections were digitized by using a stage scanner microscope (Mikroscan D2, Mikroscan Technologies, Inc., Carlsbad, CA, http://www.mikroscan.com). Histopathology of stained slides was independently evaluated by an experienced veterinary pathologist (Department of Veterinary Pathology, Mayo Clinic) who was blinded to the different treatment arms of the study. General pathological observations were provided for all H&E slides. Articular cartilage changes in the tissues of rabbits at risk of osteoarthritis (after bilateral medial anterior hemimeniscectomies) were evaluated according to the OARSI guidelines for evaluation of rabbit tissues [26, 27]. Samples in cohort B were scored for the following: staining quality of safranin O-fast green (0-6), structure of cartilage (0-11), decreases in chondrocyte density (0-4), and cluster formation (0-3). A histopathological assessment of OA synoviopathy in the right and left synovium and fat pad tissues of cohort B animals receiving injections of either (a) lactated Ringer's or (b)  $6 \times 10^{6}$  AMSCs in the right leg was also conducted according to the 0-3 scoring criteria for synoviocytes, inflammatory infiltrates, and synovial stroma parameters outlined by Laverty et al. [26].

## **Statistical Analysis**

Descriptive statistics were used to assess the effects of AMSC injection on cohort A and B animals. Mean values and SDs were reported unless otherwise indicated. A general linear model analysis was performed in SAS (Version 9.4, SAS Institute Inc., Cary, NC, http://www.sas.com) to test whether there were any significant differences between left and right leg incapacitance measurements. A repeated-measures analysis of variance (ANOVA) was performed to investigate effects related to treatment type (lactated Ringer's or AMSC) and/or leg. Reliability of the measurements was described by using intraclass correlations and coefficients of repeatability. Statistical comparisons of T2 mapping time (means) between left and right legs were assessed

by using paired *t* tests. Significant differences in right leg mapping times across treatment type were also tested by using ANOVA. In all cases, statistical significance was defined as p < .05.

## RESULTS

## Safety Evaluation of Human AMSCs After Injection Into Healthy Rabbit Knees

## Evaluation of Systemic Adverse Reactions in Healthy Rabbits

For our initial safety evaluation of human AMSC therapy within the joint space, we performed intra-articular injections of AMSCs or lactated Ringer's control into healthy knee joints of mature New Zealand white rabbits (cohort A). After injection, weight, body temperature, injection site appearance, and behavior were recorded daily. One rabbit receiving lactated Ringer's injections in both knees developed knee swelling and systemic illness consistent with septic arthritis. This rabbit was euthanized 4 days after injection. All other rabbits maintained normal behavior and stable vital signs. Excluding the rabbit that developed septic arthritis, animals were sacrificed at either 7 (n = 6) or 14 (n = 6) days after injection therapy so that intra-articular tissues could be collected and assessed for acute adverse reactions (e.g., cartilage damage).

## Assessment of Tissue Morphology/Histology in Healthy Rabbit Knees

Upon gross visual inspection, all knees appeared healthy with smooth articular cartilage. Right knees receiving experimental treatment ( $2 \times 10^6$  or  $12 \times 10^6$  cells or lactated Ringer's solution) appeared indistinguishable from the contralateral left control knees for all treatment groups (data not shown). Histologic evaluation of knee tissues including the tibia, meniscus, synovium, and fat pad, showed no evidence of cartilage injury or damage to other intra-articular joint tissues at either the day 7 or 14 time points. At 7 days, the two rabbits receiving injections of  $2 \times 10^6$ AMSCs developed lymphoplasmacytic infiltrates in their right knee; one of these rabbits also developed infiltrates in their contralateral control knee. The two rabbits receiving a dose of 12 imes10<sup>6</sup> AMSCs also developed lymphoplasmacytic infiltrates (Fig. 4). One of the rabbits developed infiltrates in both knees, whereas the other rabbit only developed lymphoplasmacytic infiltrates in the contralateral control knee. In addition, one of the rabbits receiving only lactated Ringer's therapy developed a mild lymphoplasmacytic infiltrate in the left knee, whereas the other rabbit did not develop any lymphoplasmacytic infiltrates. These data show that early lymphoplasmacytic infiltrates can be observed in rabbit knees receiving either AMSC or lactated Ringer's therapy. These cellular infiltrates could be attributable to inflammation caused by the injection procedure or the injectate itself, as well as associated joint distension.

At 14 days, all of the rabbits that received AMSC injections (at a dose of either  $2 \times 10^6$  or  $12 \times 10^6$  AMSCs) had developed lymphoplasmacytic infiltrates in their right knees, whereas their contralateral control knees were free of any cellular infiltrates. A single case of hyperplasia was present in the left synovium of one rabbit. All of the rabbits that had only received lactated Ringer's therapy were also free of lymphoplasmacytic infiltrates. These data suggest a persistent subclinical lymphoplasmacytic cellular infiltration in rabbit joints receiving human AMSC therapy. This effect may be due to an immune response

www.StemCellsTM.com



**Figure 4.** Cohort A histology. Representative H&E-stained histological samples of cohort A rabbit right knees injected with  $12 \times 10^6$  AMSCs and the corresponding control left leg (lactated Ringer's) are shown. Tissues include the femur, tibia, synovium, infrapatellar fat pad, and meniscus. (a–e): No significant pathology was observed in left leg tissues. (f–j): On the right side, mild to moderate aggregation of lymphoplasmacytic cells was observable in all tissues. No tissue injury/damage associated with the infiltrate was observed. Scale bar = 5 mm. Abbreviations: AMSC, adipose-derived mesenchymal stem cell; H&E, hematoxylin and eosin staining.

to the xenogeneic human cells. Representative histological pictures of control-treated and AMSC-treated knees from cohort A are presented in Figure 4.

## Safety Evaluation of AMSCs After Injection Into Rabbit Knees at Risk for Osteoarthritis Secondary to Partial Meniscectomy

Patients who are at risk for rapid osteoarthritis development, such as patients with meniscal injuries, are potential candidates for AMSC therapy because of its potential disease-modifying effects. We therefore wanted to evaluate the safety of AMSCs within joints at risk for osteoarthritis development, to make sure they would not contribute to rapid disease progression in highrisk joints. For these studies, we performed a bilateral medial anterior hemimeniscectomy in 18 rabbits (cohort B). Rabbits were allowed to heal for 4 weeks, at which time they were given an injection of experimental therapy (2  $\times$  10<sup>6</sup> or 6  $\times$  10<sup>6</sup> AMSCs or lactated Ringer's) into their right knee and lactated Ringer's control therapy into the left knee. Rabbits were evaluated for adverse reactions and subsequently sacrificed 4 weeks after experimental injections.

## **Evaluation of Systemic Adverse Reactions**

To assess for systemic adverse reactions of AMSC injections into rabbit joints that have undergone meniscectomy, we recorded the weight, body temperature, injection site appearance, and behavior of the rabbits weekly. One rabbit experienced a left knee infection after the bilateral partial medial meniscectomy procedure. This rabbit was sacrificed because of persistent disability at the recommendation of the veterinary staff. The sacrifice was performed before the rabbit received an experimental injection; thus, this adverse event was not related to AMSC therapy. Aside from this single rabbit, all other animals maintained normal behavior and stable vital signs. All the remaining animals in cohort B (n = 17) were sacrificed 4 weeks after injection therapy (8 weeks after partial meniscectomy) so that intra-articular tissues could be evaluated for local adverse reactions.

## Assessment of Tissue Morphology/Histology in Rabbit Knees After Partial Meniscectomy

The rabbit knees were dissected after sacrifice to assess whether injection of AMSCs into the joint had any morphological or immunological impact on the integrity of the rabbit knee. In cohort B animals, a roughened cartilage surface was evident on the medial articular surface of all tibia specimens across all treatment groups  $(2 \times 10^6 \text{ or } 6 \times 10^6 \text{ AMSCs} \text{ or lactated Ringer's solution})$  (Fig. 5). These findings are consistent with an early osteoarthritic state in a meniscetomized joint. The surface roughness that was readily identifiable upon visual inspection did not, however, translate into histologically observable measures of osteoarthritic change.

After gross examination, knee tissues, including the tibia, meniscus, synovium, and fat pad, were harvested. Representative histological pictures of control- and AMSC-treated knees are presented in Figure 6 and supplemental online Figures 1–3. Based on the OARSI scoring of the safranin O slides, the grading for

916

STEM CELLS TRANSLATIONAL MEDICINE

COHORT B



**Figure 5.** Gross morphology of articular surface after meniscectomy (cohort B). Images of rabbit tibias and femurs extracted 8 weeks after meniscectomy depicting gross morphology. (A): Left tibia, lactated Ringer's. (B): Left tibia,  $2 \times 10^6$  cells. (C): Right tibia, lactated Ringer's. (D): Right tibia,  $6 \times 10^6$  cells. (E): Right femur,  $6 \times 10^6$  cells. (F): Left femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (G): Right femur,  $6 \times 10^6$  cells. (H): Right femur,  $6 \times 10^6$  cells. (H): Right femur, lactated Ringer's. (

evaluable cartilage in all cohort B tibia samples was as follows: 0, uniform safranin O staining; 0, structure normal; 0, no decrease of cells; and 0, no cluster formation. In some samples, focal or multifocal artifactual fissures were observed, where hypocellularity or loss of safranin O staining adjacent to the cleft was present. There were no observable differences in cartilage specimens according to treatment side (left vs. right) or across animal treatment groups. Furthermore, staining provided no evidence of changes or damage to the articular cartilage.

To supplement the OARSI scoring system, a histopathological assessment of OA synoviopathy in the right and left synovium and fat pad tissues of cohort B animals receiving injections of either (a) lactated Ringer's or (b)  $6 \times 10^{6}$  AMSCs in the right leg was performed. An attempt was made to implement the 0-3 scoring criteria for synoviocytes, inflammatory infiltrates, and synovial stroma according to the parameters outlined by Laverty et al. [26]. A general pathological assessment of all H&E-stained tissues was also conducted. All of the soft tissues evaluated in cohort B received an OARSI score of 0 (normal or absent) for all relevant categories. In a few samples, abnormal histologic findings were also noted and included focal areas of necrosis, foreign body giant cells, acellular debris, and edema, which appeared to be spurious findings unrelated to the treatment that was administered. The right knees in the experimental group that were injected with higher-dose AMSC therapy showed a greater proportion of plasmalymphocytic reactions, with evidence indicative of inflammation, aggregation of lymphoplasmytic cells, and edema. All of the rabbits that received  $6 \times 10^{6}$  AMSCs developed cellular infiltrates in their right joint tissues; 50% of the rabbits receiving 2 imes 10<sup>6</sup> AMSCs developed infiltrates; and only 40% of rabbits receiving lactated Ringer's solution developed infiltrates. In the left contralateral control knees, we observed very little significant pathology; in isolated cases, mild edema or minimal aggregation of lymphoid cells was observed. In addition, necrotic debris was observed in the fat pad of an animal with lactated Ringer's and in the fat pad and synovium of two animals injected with 6 imes 10<sup>6</sup> AMSCs. As in cohort A, these studies suggest that human AMSCs may promote cellular infiltration in joint tissues, because rabbits

receiving AMSC therapy had a higher proportion of cellular infiltrations upon histologic analysis of tissues. This reaction is consistent with a normal response to xenograft AMSCs in an immunocompetent animal and did not produce a clinically significant systemic reaction or cause measurable damage to intraarticular joint tissues.

## Pain Assessment (Incapacitance Testing)

A quantitative measure of the level of discomfort (incapacitance) as it relates to injury or inflammation was collected for each rabbit in cohort B to assess whether the injection of stem cells caused the subjects to preferentially shift their weight to a specific leg secondary to pain [28, 29]. Across all rabbits, incapacitance measures (in grams) for the neutral position (N) are as follows: N<sub>Right</sub> = 1,151.4  $\pm$  173.9 g and  $N_{Left}$  = 1,089.0  $\pm$  229.0 g for the right and left side respectively, with no significant difference between legs (p = .3967). Likewise, no significant difference (p = .5835) was found between legs in the elevated (E) position:  $E_{Right}$  = 1,327.8  $\pm$ 324.9 g compared with  $E_{Left}$  = 1,280.5  $\pm$  235.6 g (supplemental online Fig. 4). Statistical analysis also revealed no differences due to treatment in the right (p = .7992) and left (p = .5237) legs or between leg and treatment type (p = .9461). The intraclass correlations (with 95% confidence intervals) and coefficients of repeatability for each leg and position are presented in supplemental online Figure 4. These findings indicate no statistically significant weight-bearing preference in any of the rabbits. Thus, the lymphoplasmacytic reactions related to AMSC therapy did not result in measurable pain behavior in our rabbit model.

# Radiographic Studies (X-Ray and Magnetic Resonance Imaging)

Bilateral AP and lateral knee x-ray radiographs were taken at the time of surgical meniscectomy, knee joint injection, and at the time of final sacrifice to evaluate for any pathologic changes in the knee joint. X-rays reveal no signs of osteoarthritic change related to AMSC therapy, as assessed by joint space narrowing, sclerosis, osteophyte formation, and joint deformity (supplemental

.com © 2016 The Authors Stem Cells Translational Medicine published by Wiley Periodicals, Inc. on behalf of AlphaMed Press



Figure 6. Cohort B histology: 6 × 10<sup>6</sup> AMSC. H&E-stained histological samples of cohort B rabbit right knees injected with 6 × 10<sup>6</sup> AMSCs and the corresponding control left legs (lactated Ringer's) are shown. Tissues include the tibia, synovium, infrapatellar fat pad, and meniscus. (a-g): No significant pathology was observed in left leg tissues (a-e) or in the right tibia samples (f, g). (h-j): The synovium (h), infrapatellar fat pad (i), and meniscus (j) had mild inflammation and aggregation of lymphohistiocytic cells, with signs of mild subintimal edema in (h) and (j). Scale bar = 5 mm. Abbreviations: AMSC, adipose-derived mesenchymal stem cell; Ant.Tibia, anterior tibial tubercle side; H&E, hematoxylin and eosin staining; Post.Tibia, posterior tibial fibular side.

online Fig. 5) stemming from joint destabilization upon partial meniscus removal.

T2 mapping by magnetic resonance imaging was performed on disarticulated rabbit legs from cohort B to develop a quantitative description of joint degeneration resulting from bilateral meniscectomies and the injection of AMSCs. Damage to articular cartilage leads to an increase in the amount of free water in the tissue, increasing T2 relaxation times [30-32]. Higher T2s are therefore reflective of abnormal or damaged cartilage, predicting early degenerative change [33, 34]. Significant increases in T2 times in milliseconds were observed for both the right medial femur (RF<sub>M</sub> = 48.5  $\pm$  9.7 milliseconds) and right medial tibia (RT<sub>M</sub> = 46.5  $\pm$  8.8 milliseconds) compared with the left medial femur (LF<sub>M</sub> = 41.7  $\pm$  8.6 milliseconds) and left medial tibia (LT<sub>M</sub> = 40.2  $\pm$  5.3 milliseconds), with p = .0037 and p = .0038, respectively (Fig. 7A). No significant differences, however, were detectable between right and left legs for different lateral tissues types: right lateral femur (RF<sub>L</sub> = 42.6  $\pm$  12.0 milliseconds) compared with the left lateral femur (LF<sub>L</sub> = 42.3  $\pm$  9.2 milliseconds), with p = .9036, and right lateral tibia (RT<sub>1</sub> = 43.3 ± 10.7 milliseconds) compared with the left lateral tibia (LT<sub>L</sub> = 43.4  $\pm$  10.0 milliseconds), with p = .9647. T2 mapping was also compared across tissues types in the right leg to test for changes due to treatment. No significant differences in T2 times were observed between

lactated Ringer's, 2 million, or 6 million AMSC-injected right legs (Fig. 7B). We conclude that, although imaging parameters do not differ for lateral tissues, partial removal of the medial meniscus likely accounts for T2 differences observed in the medial articular cartilage.

### DISCUSSION

Current treatments for OA are not able to restore normal articular cartilage integrity [3]. Because of their differentiation capacity, as well as anti-inflammatory and regenerative trophic functions, stem cells represent a novel treatment option for OA that warrants further investigation in human clinical trials [35-38]. Our focus on AMSCs as a resource is primarily driven by their ease of isolation in large quantities as a clinical-grade product that is expanded in human platelet lysate and used in autologous cell transplantation [11]. Potential risks related to the use of stem cells include colonization of nontarget tissues, possible induction or stimulation of tumorigenesis, and contamination and lack of universal protocols during isolation, as well as possible inadvertent biological complications that may arise during cell expansion [39]. The present study was designed to assess the safety and reliability of platelet-expanded human AMSCs for possible use in clinical trials to both relieve pain symptoms and/or mitigate

## COHORT B MRI



## A Difference between right and left rabbit legs according to tissue type

**Figure 7.** Rabbit magnetic resonance imaging. Magnetic resonance T2-weighted maps were generated for each rabbit knee, and the average T2 relaxation times in milliseconds for the articular cartilage of the lateral and medial femoral condyles and tibial plateaus were measured. (A): No significant differences were detectable between left and right legs for different lateral tissues (lateral femur, p = .904; lateral tibia, p = .965), whereas the right medial femur and right medial tibia had significantly higher (p = .004) T2 times compared with the left sides. (B): No significant differences in T2 were observed between lactated Ringer's or  $2 \times 10^6$  or  $6 \times 10^6$  AMSC-injected right legs (lateral femur, p = .792; lateral tibia, p = .743). \*\*, p < .005. Abbreviations: AMSC, adipose-derived mesenchymal stem cell; L, left; MRI, magnetic resonance imaging; R., right.

the progression of early stages of OA, especially in patients with acute joint injuries (e.g., meniscus tears). We examined the safety of human AMSCs after injection using two different protocols. In the first cohort (cohort A), cells were delivered into healthy rabbit knees by minimally invasive intra-articular injection, with the primary goal of evaluating for any acute adverse reactions. In the second cohort, we evaluated the response of the joint to AMSC injection 4 weeks after performing a partial anterior hemimeniscectomy. This cohort was meant to simulate a clinical scenario in which a patient sustains a meniscal injury and is predisposed to early osteoarthritis development. Future studies examining AMSC treatments at both earlier and later time points may reveal different degrees of efficacy under different clinical scenarios.

We did not observe serious adverse reactions (death, systemic reactions, or tissue damage) upon intra-articular injection when healthy rabbits maintaining normal biometric parameters received this experimental AMSC treatment. Histologic evaluation of tissues did reveal a higher incidence of cellular infiltration in knees receiving AMSC therapy. The latter response has been observed in previous animal models of OA and/or MSC studies [40–42], but it did not translate into evidence of major structural damage to cartilage or other intrasynovial tissues, as evident from the OARSI scoring for articular cartilage or synoviopathy.

Previous studies have shown that transplantation of isolated human amniotic mesenchymal cells [43], human bone marrow stromal cells (BMSCs) [40, 44], and others [42] into rabbit knee joints does not elicit apparent inflammatory responses. However, in our study we observed lymphoplasmacytic infiltration and inflammation in both cohorts. Our findings are consistent with other studies showing immune responses related to xenogeneic transplantation [42] and nonautologous transplantation of isolated cells [40, 45]. It appears there is no consensus on the biological effects of heterologous cell transplantation in rabbits.

Although OA-related joint degeneration can be mimicked in the rabbit via chemical induction, anterior cruciate ligament transection (ACLT), or meniscectomy [46–49], our studies applied the latter procedure to assess the safety of AMSCs in this more clinically relevant environment. No adverse reactions related to AMSC use were observed in this cohort, except for one animal that was removed from our study before stem cell injection due to development of a left knee infection after meniscectomy. Pain outcomes related to AMSC injections were evaluated in all animals by testing incapacitance as described in several recent studies [49-51]. We did not observe significant weight-bearing asymmetry between treatment and control groups 4 weeks after AMSC injections in cohort B, nor did we observe changes in spontaneous locomotor activities or gait, suggesting that cell treatment neither contributes to nor alleviates pain in the rabbit hind limbs at this time point. In this study, we limited incapacitance testing to terminal time points just before euthanasia to avoid stressing the rabbits, because the rabbits often became very agitated during handling. Future assessments of knee pain with incapacitance testing using more frequent, as well as earlier and later, time points may reveal significant pain-related findings.

As anticipated, gross morphological inspection of the joints after sacrifice suggested early signs of cartilage dysregulation, evidenced by the presence of mild fibrillation of the articular cartilage. Radiological examination of these same joints did not reveal major areas of joint space narrowing, sclerosis, or osteophyte formation, and we did not find indications for structural cartilage degeneration by H&E staining or safranin O-fast green staining. Hence, the meniscectomy procedure carried out in our study generated mild preclinical OA symptoms that were far less pronounced than the radiological and/or histological signs of OA that have been previously reported at 8–24 weeks after surgery in models of ACLT [17, 36, 52] and meniscal injury [35, 53, 54].

This safety study also provides additional preliminary information regarding cartilage damage by measurements of MR relaxation time T2. Degeneration changes have been shown to increase T2 times in human articular cartilage [55], particularly when comparing healthy and OA-damaged subjects [56]. Although we did observe increases in T2 in the right medial knee tissues compared with their left counterparts, no significant differences were found across treatment types, suggesting that any measureable changes in cartilage structure were not due to AMSC dose alone. Because T2 signals are highly sensitive to various factors (e.g., background, orientation) and there is the potential for substantial heterogeneity within healthy cartilage, interpretation of these results is challenging [57]. Additional studies are currently ongoing to fully realize the clinical values of these types of radiographic parameters.

Our findings indicate that, aside from the presence of cellular infiltrates in the rabbit joint tissues, there is minimal risk associated with AMSC injection into the animals at the doses we used  $(2 \times 10^{6} \text{ to } 12 \times 10^{6})$ . This study contributes to the growing evidence that MSCs can be used safely in vivo for the duration of our experiments (up to 4 weeks). In our study, we focused on relatively short time points (2-4 weeks) to ensure that we would be able to capture acute adverse reactions directly related to the AMSC therapy. However, studies examining longer time points after AMSC injections, as well as multiple dosing regimens with autologous cell therapies, would be interesting, particularly given that recent studies have shown that MSCs may have the potential to persist within the joint space and joint tissues for longer periods of time after intra-articular administration [58]. A number of preclinical studies evaluating the regenerative potential of BMSCs, AMSCs, and other types of MSCs after meniscus injury provide evidence of tissue regeneration, protection against joint surface irregularities, and improvement in cartilage quality over controls [35, 36, 53, 54]. This response is potentially because of trophic mechanisms of action, via the release of growth factors and cytokines, MSC adherence and differentiation, and extracellular matrix deposition [17, 18], although more rigorous studies on the

influence of catabolic and inflammatory environment during OA are necessary to elucidate the specific mechanisms involved. The safety and efficacy of intra-articular injections of bone marrow-derived MSCs has been evaluated in patients with knee OA [59-61], with no measured adverse events and reports of pain reduction and improved function. In a recent publication, Vangsness et al. [18] followed patients in a phase I clinical study for 2 years, with good demonstration of MSC safety and relative improvements in pain. It remains unclear, however, whether clinically relevant increases in meniscal regeneration are possible. Improvements in areas such as dose optimization, standardization of outcome measures, evaluation of the full joint environment, stem cell tracking and survival, and the feasibility of repeat injections will need to be explored further to balance OA treatment with the safe delivery of MSCs.

Although our results suggest that human AMSC injections are well tolerated in the rabbit model, a number of limitations still need to be addressed. Although the rabbit is a practical model for early stages of evaluation because of cost-effectiveness, handling, and reasonable joint size [62], rabbits are associated with spontaneous intrinsic healing of cartilage—a property generally not seen in humans [46, 52, 63, 64]—and have distinctly reduced meniscus vascularization (compared with humans) [65], which could significantly alter the disease environment and influence outcomes. A longer follow-up period, with the potential to track cells after injection, could provide more detailed information about cell fate, cell impact, and OA induction. An area of ongoing investigation in the field of translational medicine is whether human cells that will ultimately be administered to patients or cells derived from the animal model under study should be used for safety investigations. In this study, a safety evaluation of human-derived cells applied to a rabbit model of OA proved sufficient for regulatory approval in the setting of prior experience with MSC therapy. However, there is still a strong need for future investigations to explore the differences between MSCs across species to understand what factors are important for safety and efficacy evaluations and to determine which aspects can help optimize cell-based manufacturing practices. In our studies, we observed immune reactions associated with xenogeneic AMSC therapy. Previous experiences with the use of autologous AMSC therapies in patients, for the treatment of diseases outside of osteoarthritis. have not shown evidence of adverse immune reactions [66, 67]. However, without these prior human studies, many of which have been for diseases associated with high morbidity that lack effective therapies, the translational barriers for more benign diseases such as OA can be challenging to overcome, given the state of current translational models for cell-based therapies.

## CONCLUSION

We present the results of a study evaluating the safety of human adipose-derived mesenchymal stromal/stem cells delivered as single intra-articular injections (at three dose levels) into normal healthy rabbit knees and into knees at risk for the development of osteoarthritis after bilateral medial meniscectomy. Overall, study injections were well tolerated up to 4 weeks after injection, demonstrating no major adverse effects on tests of general health, function and behavior of animals. Likewise, we did not find imaging (radiographic or MR), gross anatomic, or histologic evidence postmortem of concerning risk in this study. The results suggest

## ACKNOWLEDGMENTS

We thank the members of our research group, including Paul Siegert, Markus Schreiner, Endre Soreide, and Roman Thaler, for general support of this project, as well as Jennifer Westendorf, David Devle, and Noelle Larson for stimulating discussions. We also thank Kristin C. Mara for assistance with statistical analysis and Ron Marler (Mayo Clinic, Scottsdale, AZ) for pathological interpretations. This work was supported, in whole or in part, by National Institutes of Health Grants R01 AR049069 (A.J.v.W.), T32 AR056950 (E.A.L., Principal Investigator: Jennifer Westendorf), and F32 AR066508 (A.B.D. and A.J.v.W.). This work has also been supported by the Mayo Graduate School, Clinical and Translational Sciences Track. Additional support was provided by the Center for Clinical and Translational Science (UL1 TR000135) and intramural grants from the Center for Regenerative Medicine at Mayo Clinic. We also appreciate the generous philanthropic support of William H. and Karen J. Eby and the charitable foundation in their names.

#### 921

#### **AUTHOR CONTRIBUTIONS**

S.M.R.: conception and design, collection and/or assembly of data, data analysis and interpretation, manuscript writing, final approval of manuscript; J.M.D., D.L.J., and J.S.: collection and/or assembly of data, data analysis and interpretation, manuscript writing, final approval of manuscript; Y.L. and M.A.F.: collection and/or assembly of data, data analysis and interpretation, final approval of manuscript; T.d.M., E.A.L., H.N., C.R.P., D.J.R., A.D., and E.T.C.: collection and/or assembly of data, data analysis and interpretation, final approval of manuscript; D.R.L., W.Q., A.J.K., and H.-J.I.: data analysis and interpretation, final approval of manuscript; A.B.D.: provision of study material or patients, data analysis and interpretation, final approval of manuscript; A.J.v.W.: conception and design, financial support, data analysis and interpretation, manuscript writing, final approval of manuscript.

## DISCLOSURE OF POTENTIAL CONFLICTS OF INTEREST

A.B.D. is on the board of directors and has stock options with Mill Creek Life Sciences and has rights to royalties. J.S. is an employee of Sonex Health, LLC, and has intellectual property rights and stock options with Tenex Health, Inc., and Sonex Health, LLC. The other authors indicated no potential conflicts of interest.

#### REFERENCES

**1** Maradit Kremers H, Larson DR, Crowson CS et al. Prevalence of total hip and knee replacement in the United States. J Bone Joint Surg Am 2015;97:1386–1397.

2 Kurtz S, Ong K, Lau E et al. Projections of primary and revision hip and knee arthroplasty in the United States from 2005 to 2030. J Bone Joint Surg Am 2007;89:780–785.

**3** Mastbergen SC, Saris DB, Lafeber FP. Functional articular cartilage repair: Here, near, or is the best approach not yet clear? Nat Rev Rheumatol 2013;9:277–290.

4 Farkouh ME, Kirshner H, Harrington RA et al. Comparison of lumiracoxib with naproxen and ibuprofen in the Therapeutic Arthritis Research and Gastrointestinal Event Trial (TARGET), cardiovascular outcomes: randomised controlled trial. Lancet 2004;364:675–684.

**5** Tammachote N, Kanitnate S, Yakumpor T et al. Intra-articular, single-shot Hylan G-F 20 hyaluronic acid injection compared with corticosteroid in knee osteoarthritis: A double-blind, randomized controlled trial. J Bone Joint Surg Am 2016;98:885–892.

**6** Filardo G, Di Matteo B, Di Martino A et al. Platelet-rich plasma intra-articular knee injections show no superiority versus viscosupplementation: A randomized controlled trial. Am J Sports Med 2015;43:1575–1582.

**7** Wagner BS, Howe AS, Dexter WW et al. Tolerability and efficacy of 3 approaches to intra-articular corticosteroid injections of the knee for osteoarthritis: A randomized controlled trial. Orthop J Sports Med 2015; 3:2325967115600687.

**8** Fubini SL, Todhunter RJ, Burton-Wurster N et al. Corticosteroids alter the differentiated phenotype of articular chondrocytes. J Orthop Res 2001;19:688–695.

**9** Dudakovic A, Camilleri E, Riester SM et al. High-resolution molecular validation of

self-renewal and spontaneous differentiation in clinical-grade adipose-tissue derived human mesenchymal stem cells. J Cell Biochem 2014; 115:1816–1828.

**10** Dudakovic A, Camilleri ET, Lewallen EA et al. Histone deacetylase inhibition destabilizes the multi-potent state of uncommitted adipose-derived mesenchymal stromal cells. J Cell Physiol 2015;230:52–62.

**11** Crespo-Diaz R, Behfar A, Butler GW et al. Platelet lysate consisting of a natural repair proteome supports human mesenchymal stem cell proliferation and chromosomal stability. Cell Transplant 2011;20:797–811.

12 Minteer D, Marra KG, Rubin JP. Adiposederived mesenchymal stem cells: biology and potential applications. Adv Biochem Eng Biotechnol 2013;129:59–71.

**13** Dave M, Mehta K, Luther J et al. Mesenchymal stem cell therapy for inflammatory bowel disease: A systematic review and meta-analysis. Inflamm Bowel Dis 2015;21:2696–2707.

**14** Carceller MC, Guillén MI, Ferrándiz ML et al. Paracrine in vivo inhibitory effects of adipose tissue-derived mesenchymal stromal cells in the early stages of the acute inflammatory response. Cytotherapy 2015;17:1230–1239.

**15** Loeser RF, Collins JA, Diekman BO. Ageing and the pathogenesis of osteoarthritis. Nat Rev Rheumatol 2016;12:412–420.

**16** Bourin P, Bunnell BA, Casteilla L et al. Stromal cells from the adipose tissue-derived stromal vascular fraction and culture expanded adipose tissue-derived stromal/stem cells: A joint statement of the International Federation for Adipose Therapeutics and Science (IFATS) and the International Society for Cellular Therapy (ISCT). Cytotherapy 2013;15:641–648.

**17** Desando G, Cavallo C, Sartoni F et al. Intra-articular delivery of adipose derived stromal cells attenuates osteoarthritis progression in an experimental rabbit model. Arthritis Res Ther 2013;15:R22. **18** Vangsness CT Jr., Farr J 2nd, Boyd J et al. Adult human mesenchymal stem cells delivered via intra-articular injection to the knee following partial medial meniscectomy: A randomized, double-blind, controlled study. J Bone Joint Surg Am 2014;96:90–98.

**19** Lindhorst E, Wachsmuth L, Kimmig N et al. Increase in degraded collagen type II in synovial fluid early in the rabbit meniscectomy model of osteoarthritis. Osteoarthritis Cartilage 2005;13:139–145.

**20** Colombo C, Butler M, O'Byrne E et al. A new model of osteoarthritis in rabbits. I. Development of knee joint pathology following lateral meniscectomy and section of the fibular collateral and sesamoid ligaments. Arthritis Rheum 1983;26:875–886.

**21** Gao J, Messner K. Natural healing of anterior and posterior attachments of the rabbit meniscus. Clin Orthop Relat Res 1996(328):276–284.

22 Anetzberger H, Mayer A, Glaser C et al. Meniscectomy leads to early changes in the mineralization distribution of subchondral bone plate. Knee Surg Sports Traumatol Arthrosc 2014; 22:112–119.

**23** Messner K, Fahlgren A, Persliden J et al. Radiographic joint space narrowing and histologic changes in a rabbit meniscectomy model of early knee osteoarthrosis. Am J Sports Med 2001;29:151–160.

**24** Olson S, Guilak F. Post-Traumatic Arthritis: Pathogenesis, Diagnosis and Management. New York, NY: Springer, 2015:360.

**25** Martin JT, Collins CM, Ikuta K et al. Population average T2 MRI maps reveal quantitative regional transformations in the degenerating rabbit intervertebral disc that vary by lumbar level. J Orthop Res 2015;33:140–148.

26 Laverty S, Girard CA, Williams JM et al. The OARSI histopathology initiative—recommendations for histological assessments of osteoarthritis in the rabbit. Osteoarthritis Cartilage 2010;18 (suppl 3):S53–S65.

27 Pritzker KP, Gay S, Jimenez SA et al. Osteoarthritis cartilage histopathology: Grading and staging. Osteoarthritis Cartilage 2006;14:13-29.

28 Columbus Instruments Incapacitance tester: Incap. Columbus, OH. Available at http:// www.colinst.com/brief.php?id=22. Accessed August 18. 2015.

29 Bioseb-In Vivo Research Instruments. Static weight bearing: Incapacitance test. Pinellas Park. Available at http://www.bioseb.com/bioseb/ anglais/default/item\_id=27.php. Accessed August 18, 2015.

30 Maroudas A, Venn M. Chemical composition and swelling of normal and osteoarthrotic femoral head cartilage. II. Swelling. Ann Rheum Dis 1977:36:399-406.

31 Maroudas AI. Balance between swelling pressure and collagen tension in normal and degenerate cartilage. Nature 1976;260:808-809.

32 Watrin-Pinzano A. Ruaud JP. Cheli Y et al. T2 mapping: An efficient MR quantitative technique to evaluate spontaneous cartilage repair in rat patella. Osteoarthritis Cartilage 2004;12: 191-200.

33 David-Vaudey E, Ghosh S, Ries M et al. T2 relaxation time measurements in osteoarthritis. Magn Reson Imaging 2004;22:673-682.

34 Mosher TJ, Dardzinski BJ. Cartilage MRI T2 relaxation time mapping: Overview and applications. Semin Musculoskelet Radiol 2004; 8:355-368.

35 Shen W, Chen J, Zhu T et al. Osteoarthritis prevention through meniscal regeneration induced by intra-articular injection of meniscus stem cells. Stem Cells Dev 2013;22:2071-2082.

36 Singh A, Goel SC, Gupta KK et al. The role of stem cells in osteoarthritis: An experimental study in rabbits. Bone Joint Res 2014;3:32-37.

37 Toghraie FS, Chenari N, Gholipour MA et al. Treatment of osteoarthritis with infrapatellar fat pad derived mesenchymal stem cells in rabbit. Knee 2011;18:71-75.

38 ter Huurne M, Schelbergen R, Blattes R et al. Antiinflammatory and chondroprotective effects of intraarticular injection of adiposederived stem cells in experimental osteoarthritis. Arthritis Rheum 2012:64:3604-3613

39 Lim MH, Ong WK, Sugii S. The current landscape of adipose-derived stem cells in clinical applications. Expert Rev Mol Med 2014;16: e8.

40 Jang KM, Lee JH, Park CM et al. Xenotransplantation of human mesenchymal stem cells for repair of osteochondral defects in rabbits using osteochondral biphasic composite constructs. Knee Surg Sports Traumatol Arthrosc 2014:22:1434-1444

41 Pei M, Yan Z, Shoukry M et al. Failure of xenoimplantation using porcine synoviumderived stem cell-based cartilage tissue constructs for the repair of rabbit osteochondral defects. J Orthop Res 2010;28:1064-1070.

42 Saulnier N, Viguier E, Perrier-Groult E et al. Intra-articular administration of xenogeneic neonatal mesenchymal stromal cells early after meniscal injury down-regulates metalloproteinase gene expression in synovium and prevents cartilage degradation in a rabbit model of osteoarthritis. Osteoarthritis Cartilage 2015;23:122-133.

43 Nogami M, Tsuno H, Koike C et al. Isolation and characterization of human amniotic mesenchymal stem cells and their chondrogenic differentiation. Transplantation 2012; 93:1221-1228

44 Hong JH, Park JI, Kim KH et al. Repair of the complete radial tear of the anterior horn of the medial meniscus in rabbits: A comparison between simple pullout repair and pullout repair with human bone marrow stem cell implantation. Knee Surg Relat Res 2011;23:164–170.

45 Revell CM, Athanasiou KA. Success rates and immunologic responses of autogenic, allogenic, and xenogenic treatments to repair articular cartilage defects. Tissue Eng Part B Rev 2009:15:1-15

46 Gregory MH, Capito N, Kuroki K et al. A review of translational animal models for knee osteoarthritis. Arthritis 2012:2012:764621.

47 Dinser R. Animal models for arthritis. Best Pract Res Clin Rheumatol 2008:22:253-267.

48 Brophy RH, Martinez M, Borrelli J Jr et al. Effect of combined traumatic impact and radial transection of medial meniscus on knee articular cartilage in a rabbit in vivo model. Arthroscopy 2012;28:1490-1496.

49 Dong J, Jiang D, Wang Z et al. Intraarticular delivery of liposomal celecoxibhyaluronate combination for the treatment of osteoarthritis in rabbit model. Int J Pharm 2013: 441:285-290.

50 Mihara M, Higo S, Uchiyama Y et al. Different effects of high molecular weight sodium hyaluronate and NSAID on the progression of the cartilage degeneration in rabbit OA model. Osteoarthritis Cartilage 2007;15:543-549.

51 Ito S. Sato M. Yamato M et al. Repair of articular cartilage defect with layered chondrocvte sheets and cultured synovial cells. Biomaterials 2012;33:5278-5286.

52 Kim SS, Kang MS, Lee KY et al. Therapeutic effects of mesenchymal stem cells and hyaluronic acid injection on osteochondral defects in rabbits' knees. Knee Surg Relat Res 2012;24: 164-172

53 Horie M, Driscoll MD, Sampson HW et al. Implantation of allogenic synovial stem cells promotes meniscal regeneration in a rabbit meniscal defect model. J Bone Joint Surg Am 2012:94:701-712

54 Ruiz-Ibán MA, Díaz-Heredia J, García-Gómez I et al. The effect of the addition of adipose-derived mesenchymal stem cells to a meniscal repair in the avascular zone: An experimental study in rabbits. Arthroscopy 2011;27: 1688-1696

55 Dardzinski BJ, Mosher TJ, Li S et al. Spatial variation of T2 in human articular cartilage. Radiology 1997;205:546-550.

56 Mosher TJ, Dardzinski BJ, Smith MB. Human articular cartilage: Influence of aging and early symptomatic degeneration on the spatial variation of T2-preliminary findings at 3 T. Radiology 2000;214:259-266.

57 Wang YX. In vivo magnetic resonance imaging of animal models of knee osteoarthritis. Lab Anim 2008;42:246-264.

58 Toupet K, Maumus M, Peyrafitte JA et al. Long-term detection of human adipose-derived mesenchymal stem cells after intraarticular injection in SCID mice. Arthritis Rheum 2013;65: 1786-1794

59 Jo CH, Lee YG, Shin WH et al. Intraarticular injection of mesenchymal stem cells for the treatment of osteoarthritis of the knee: A proof-of-concept clinical trial. STEM CELLS 2014;32:1254-1266.

60 Koh YG, Choi YJ. Infrapatellar fat padderived mesenchymal stem cell therapy for knee osteoarthritis. Knee 2012;19:902-907.

61 Koh YG, Jo SB, Kwon OR et al. Mesenchymal stem cell injections improve symptoms of knee osteoarthritis. Arthroscopy 2013;29: 748-755.

62 Chu CR, Szczodry M, Bruno S. Animal models for cartilage regeneration and repair. Tissue Eng Part B Rev 2010;16:105-115.

63 Little CB, Zaki S. What constitutes an "animal model of osteoarthritis"-the need for consensus? Osteoarthritis Cartilage 2012;20: 261-267.

64 Cook JL, Hung CT, Kuroki K et al. Animal models of cartilage repair. Bone Joint Res 2014;3:89-94.

65 Chevrier A, Nelea M, Hurtig MB et al. Meniscus structure in human, sheep, and rabbit for animal models of meniscus repair. J Orthop Res 2009:27:1197-1203.

66 Lightner AL, Dozois E, Fletcher JG et al. Early results using an adipose derived mesenchymal stem cells coated fistula plug for the treatment of refractory perianal fistulizing Crohns disease. Gastroenterology 2016;150: S483-S484.

67 Behfar A, Crespo-Diaz R, Terzic A et al. Cell therapy for cardiac repair-lessons from clinical trials. Nat Rev Cardiol 2014;11:232-246.

See www.StemCellsTM.com for supporting information available online.